ORDINARY AND EXTRAORDINARY SHAREHOLDERS' MEETING

File: TC/CL/2200669/lv Repertorium: 2020/0436

"Anheuser-Busch InBev"

limited liability company
in its capacity of listed company
at 1000 Brussels, Grote Markt 1
Company number 0417.497.106
Register of Legal Entities Brussels, Dutch-language section

<u>www.ab-inbev.com</u> corporategovernance@ab-inbev.com

AMENDMENT TO THE RULES OF WRITTEN DECISION-MAKING IN THE BOARD OF DIRECTORS

--

POWER TO DISTRIBUTE INTERIM DIVIDENDS

--

ADAPTATION OF THE ARTICLES OF ASSOCIATION TO THE BELGIAN CODE OF COMPANIES AND ASSOCIATIONS

APPROVAL OF ANNUAL ACCOUNTS

--

DISCHARGES

--

APPOINTMENT DIRECTORS

--

REMUNERATION POLICY AND REMUNERATION REPORT

Today on 3 June 2020.

At 3000 Leuven, Brouwerijplein 1.

Before **Philippe MISSOUL**, **notary in Bierbeek**, with the intervention of **Tim CARNEWAL**, Associated Notary in Brussels (first district), carrying out his profession in the company "Berquin Notarissen", having its registered office at Brussels, Lloyd Georgelaan 11, not competent *ratione loci*.

WAS HELD

the ordinary and extraordinary shareholders' meeting of the listed limited liability company "Anheuser-Busch InBev", having the capacity of a listed company, with registered office at 1000 Brussels, Grote Markt 1, hereinafter the "Company" or "AB InBev".

IDENTIFICATION OF THE COMPANY

The Company was incorporated as "Newbelco" by a deed established by Mr. Peter Van Melkebeke, Notary in Brussels, on 3 March 2016, published in the Annexes of the Belgian State Gazette of 8 March 2016, under numbers 16305365 and 16305366, and rectified by deed established by Mr. Tim Carnewal, notary in Brussels, on 20 June 2016, published in the Annexes of the Belgian State Gazette of 4 July 2016, under number 16092438.

The articles of association have been amended several times and for the last time pursuant to a notarial deed passed by Mr. Tim Carnewal, Notary in Brussels, on 24 April 2019, published in

the Annexes of the Belgian State Gazette of 14 May 2019, under the numbers 19064982 and 19064983.

The Company is registered with the register of legal entities under number 0417.497.106.

OPENING OF THE MEETING - COMPOSITION OF THE BUREAU

The meeting opens at 2:00 pm, under the chairmanship of Mr. **BARRINGTON Martin J.**, who participates in the meeting by videoconference.

Composition of the bureau

The chairman indicates that, in accordance with Article 36 of the articles of association, he has constituted the bureau of the meeting on this 3 June 2020, prior to the opening of the meeting, by proceeding with the appointment of Mr. **VANDERMEERSCH Jan Dirk**, residing at 1850 Grimbergen, Prinsenstraat 55, as secretary.

Were appointed as tellers:

- Ann RANDON
- Ingvild VAN LYSEBETTEN.

VERIFICATIONS BY THE BUREAU

Preliminary statement concerning

the current exceptional circumstances (Covid-19 Crisis)

The Chairman explains that, in view of the current crisis (Covid-19 crisis), in accordance with Article 6 of Royal Decree No. 4 of 9 April 2020 containing various provisions on co-ownership and company and association law in the context of the fight against the Covid-19 pandemic (hereinafter referred to as the "Royal Decree No. 4"), the Company has decided that the general meeting will be held without the physical attendance of the shareholders.

Furthermore, the shareholders can only vote remotely (voting by correspondence) ahead of the general meeting or by proxy, which must contain specific voting instructions for each proposed resolution to be granted to Mr. Jan Vandermeersch.

The forms for vote by correspondence and the forms for proxy were made available to shareholders on the Company's website (www.ab-inbev.com).

The valid proxies with voting instructions given to a person other than Mr Jan Vandermeersch, referred to above, will be taken into account, even if the proxy holder, included in the proxy, is not physically present at this general meeting.

The chairman declares that the Company only allowed written questions in accordance with Article 6, §3 of Royal Decree No. 4, as the meeting was held without the physical attendance of the shareholders. The answers to the questions will be given orally at this general meeting which will be recorded.

The recording will be made available on the Company's website.

The chairman reports to the meeting on the findings and verifications made by the bureau during and after the formalities for the registration of participants, with a view to the constitution of the meeting.

I. Notice to the holders of securities

Prior to the opening of the meeting, the proof of convening notices published in the *Belgian State Gazette* and in the press were submitted to the bureau. The bureau acknowledged that the dates of the publications are the following:

- on 28 April 2020 in the Belgian State Gazette;
- on 28 April 2020 in De Tijd (Dutch), and L'Echo (French).

A notice was sent to several press agencies to ensure international distribution.

The text of the convening notice as well as the proxy forms and the vote by correspondence forms were also made available to the shareholders on the website of the Company (www.ab-inbev.com) as from 28 April 2020. An adapted version of documents incorporating the organisational measures taken by Royal Decree No. 4 was made available on the Company's website on 19 May 2020.

The bureau has also acknowledged that a convening notice was sent to the holders of registered securities in accordance with Articles 3:35 and 7:128 of the Belgian Code of Companies and Associations, as well as to the Directors and the statutory auditor.

II. Verification of the powers of the participants of the meeting

With respect to the participation in the extraordinary shareholders' meeting, the bureau verified whether Article 33 of the articles of association was complied with. The bureau confirmed this to the notary and the various documents evidencing so as well as the original proxies will be kept in the Company's files.

In derogation of the procedures mentioned in the original notice convening the general meeting, in accordance with the aforementioned measures taken by the Company in application of Article 6 of Royal Decree No. 4, the signed proxies, votes by correspondence as well as any written questions only had to reach the Company no later than the fourth day prior to the day of the general meeting, i.e. no later than 30 May 2020.

Compliance with the aforementioned formalities has been confirmed to the notary by the bureau. The various documents evidencing so as well as the proxies and votes by correspondence, of which a scanned or photographed version will suffice, will be kept in the Company's files.

III. Verification of the presence quorum

An attendance sheet was drawn up. This list was signed by the secretary of the meeting. This list has been supplemented with a list of all shareholders who voted by correspondence in accordance with Article 35 of the articles of association.

The bureau has acknowledged that, on the basis of the attendance sheet, the shareholders present or represented at the meeting hold 1,521,581,699 shares out of a total of 2,019,241,973 shares issued by the Company.

However the Company holds 46,773,750 own shares, so that the voting rights attached to these shares are suspended and are not taken into account to establish the presence and majority requirements to be complied with by the shareholders' meeting, in accordance with Article 7:217, §1 juncto Article 7:140 the Belgian Code of Companies and Associations. Consequently only a total of 1,972,468,223 shares issued by the Company must be taken into account.

As a result, the bureau has determined that the meeting can validly deliberate on the items on the agenda.

IV. Other parties attending the meeting

In addition to the members of the bureau and the directors who cannot participate in the meeting remotely, no persons are allowed to be physically present at this general meeting, pursuant to the decision of the Company, taken in application of article 6 of the Royal Decree No. 4

COMPOSITION OF THE MEETING - ATTENDANCE LIST

The chairman subsequently invites the shareholders' meeting to acknowledge that it is validly constituted.

Are represented the shareholders whose surname, first name and address or legal form, name and registered office, as well as the number of shares held by each of them are mentioned on the attendance list which will remain attached to the present deed.

This list has been signed by the shareholders' proxy. The secretary hands the proxies to the notary, who keeps them in his file.

Subsequently, the attendance list was provided by me notary with the notification "annex" and closed by the signature of the members of the bureau and the undersigned notary.

AGENDA

The chairman reminds that the agenda of the meeting is the following:

A. RESOLUTIONS WHICH CAN BE VALIDLY ADOPTED IF THE SHAREHOLDERS PRESENT OR REPRESENTED AT THE MEETING REPRESENT AT LEAST HALF OF THE CAPITAL, SUBJECT TO THE APPROVAL BY AT LEAST 75% OF THE VOTES CAST

1. Revised text of articles of association

- a. *Proposed resolution*: amending article 24.4 of the articles of association in order to allow the Board to decide by way of written resolutions under the conditions of the Belgian Code of Companies and Associations, as follows:
- "24.4 Any or all of the directors may participate in a meeting of the Board of Directors by means of telephone, videoconference or similar communications equipment by means of which all persons participating in the meeting can hear each other. Participation in a meeting by such means shall constitute presence in person at such meeting. Decisions of the Board of Directors may also be adopted, without any physical meeting, by the unanimous consent of the directors expressed in writing."
- b. *Proposed resolution*: amending article 44 of the articles of association in order to allow the Board to distribute interim dividends under the conditions of the Belgian Code of Companies and Associations, as follows:

"The annual dividends shall be paid at the dates and places decided by the Board of Directors.

The Board of Directors may pay an interim dividend in accordance with article 7:213 of the Code."

- c. *Proposed resolution*: amending several provisions of the articles of association in order to conform such provisions with the changes imposed by or resulting from the rules of the Belgian Code of Companies and Associations and other Belgian legislations and to align the text of the articles of association to the terminology and numbering of such Code. The proposed revised text of articles of association is available on the Company's website as indicated in this notice.
- B. RESOLUTIONS WHICH CAN BE VALIDLY ADOPTED IRRESPECTIVE OF THE CAPITAL REPRESENTED BY THE SHAREHOLDERS PRESENT OR REPRESENTED AT THE MEETING, SUBJECT TO THE APPROVAL BY AT LEAST THE MAJORITY OF THE VOTES CAST
- **2. Management report** by the Board of Directors on the accounting year ended on 31 December 2019.
- **3. Report by the statutory auditor** on the accounting year ended on 31 December 2019.
- 4. Communication of the consolidated annual accounts relating to the accounting year ended on 31 December 2019, as well as the management report by the Board of Directors and the report by the statutory auditor on the consolidated annual accounts.

Approval of the statutory annual accounts

Proposed resolution: approving the statutory annual accounts relating to the accounting year ended on 31 December 2019, including the following allocation of the result:

Profit of the accounting year: + 10,404,860
Profit carried forward: + 12,604,362

Result to be allocated: = 23,009,222

Transfer from reserves: + 0

Deduction for the unavailable reserve: - 791,390

Gross dividend for the shares (*): - 2,556,697

Balance of carried forward profit: = 19,661,136

(*) On a per share basis, this represents a gross dividend for 2019 of EUR 1.30.

Taking into account the gross interim dividend of EUR 0.80 per share paid in November 2019, a balance gross amount of EUR 0.50 will be payable as from 11 June 2020 (JSE: 15 June 2020), i.e. a balance dividend net of Belgian withholding tax of EUR 0.35 per share (in case of 30% Belgian withholding tax) and of EUR 0.50 per share (in case of exemption from Belgian withholding tax).

The actual gross dividend amount (and, subsequently, the balance amount) may fluctuate depending on possible changes in the number of own shares held by the Company on the dividend payment date.

6. Discharge to the directors

Proposed resolution: granting discharge to the directors for the performance of their duties during the accounting year ended on 31 December 2019.

7. Discharge to the statutory auditor

Proposed resolution: granting discharge to the statutory auditor for the performance of his duties during the accounting year ended on 31 December 2019.

8. Appointment of directors

- a. *Proposed resolution:* upon proposal from the Board of Directors, renewing the appointment of <u>Ms. Michele Burns</u> as independent director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Ms. Burns expressly stated and the Board is of the opinion that she complies with the functional, family and financial criteria of independence as provided for in the 2020 Belgian Corporate Governance Code.
- b. *Proposed resolution:* upon proposal from the Board of Directors, renewing the appointment of **Mr. Elio Leoni Sceti** as independent director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Mr. Sceti expressly stated and the Board is of the opinion that he complies with the functional, family and financial criteria of independence as provided for in the 2020 Belgian Corporate Governance Code.
- c. *Proposed resolution:* upon proposal from the Reference Shareholder, renewing the appointment as director of **Mr. Alexandre Van Damme**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.
- d. *Proposed resolution:* upon proposal from the Reference Shareholder, renewing the appointment as director of **Mr. Grégoire de Spoelberch**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.
- e. *Proposed resolution:* upon proposal from the Reference Shareholder, renewing the appointment as director of <u>Mr. Paul Cornet de Ways Ruart</u>, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.
- f. *Proposed resolution:* upon proposal from the Reference Shareholder, renewing the appointment as director of **Mr. Paulo Lemann**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.
- g. *Proposed resolution:* upon proposal from the Reference Shareholder, renewing the appointment as director of **Ms. María Asuncion Aramburuzabala**, for a period of four years ending

at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

- h. Proposed resolution: acknowledging the end of the mandate of Mr. Marcel Herrmann Telles as director and, upon proposal from the Reference Shareholder, appointing Mr. Roberto Thompson Motta as director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Mr. Roberto Thompson Motta, a Brazilian citizen, received a BS in Mechanical Engineering from Pontifícia Universidade Católica do Rio de Janeiro and an MBA from The Wharton School of the University of Pennsylvania. He is a co-founder and member of the Investment Committee of 3G Capital, a global investment firm headquartered in New York. Mr. Thompson served on the Board of Directors of AB InBev from 2004 until 2014 and has served since 2001 on the Board of Directors of AmBev S.A. where he is a member of the Operational and Finance Committee. Mr. Thompson has served on the Board of Directors of Restaurant Brands International since 2013, Lojas Americanas S.A. since 2001, São Carlos Empreendimentos e Participações S.A. since 2001 and StoneCo Ltd., a leading provider of financial technology solutions, since 2018, where he chairs the Finance Committee. From 1993 to 2004, he was one of the founding partners of GP Investments Ltd. and a member of its Board of Directors until 2010. Mr. Thompson is a member of the Academy of the University of Pennsylvania, The International Council of The Metropolitan Museum of Art in New York and a Patron of the Museum of Modern Art of São Paulo.
- i. *Proposed resolution:* upon proposal from the Restricted Shareholders, renewing the appointment as Restricted Share Director of **Mr. Martin J. Barrington**, for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.
- j. *Proposed resolution:* upon proposal from the Restricted Shareholders, renewing the appointment as Restricted Share Director of **Mr. William F. Gifford, Jr.**, for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.
- k. *Proposed resolution:* upon proposal from the Restricted Shareholders, renewing the appointment as Restricted Share Director of <u>Mr. Alejandro Santo Domingo Dávila</u>, for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.

9. Remuneration policy and remuneration report

Proposed resolution: approving the remuneration report for the financial year 2019 as set out in the 2019 annual report, including the remuneration policy. The 2019 annual report and remuneration report containing the remuneration policy are available on the Company's website as indicated in this notice.

C. POWERS

10. Filings

Proposed resolution: Without prejudice to other delegations of powers to the extent applicable, granting powers to Jan Vandermeersch, Global Legal Director Corporate, with power to substitute, to proceed to (i) the signing of the restated articles of association and their filings with the clerk's office of the Enterprise Court of Brussels as a result of the approval of the resolutions listed under item 1 above, and (ii) any other filings and publication formalities in relation to the above resolutions.

PRESENTATION BY MR. CARLOS BRITO AND MR. MARTIN BARRINGTON

Subsequently, Mr. Carlos Brito gives a presentation explaining the consolidated financial

results of the Company for 2019, as published in the Company's annual report, the First Quarter 2020 results and a short business update in view of the Covid-19 pandemic. After this presentation, Mr. Martin J. BARRINGTON provides comments on the remuneration report 2019 which is included in the Annual Report 2019 and which is submitted to the annual shareholders' meeting for approval.

The detailed contents of these presentations are not recorded in these minutes.

VOTING MODALITIES

The chairman subsequently invites the participants to proceed to the voting on each of the proposed resolutions on the agenda.

The chairman reminds the meeting that each share gives the right to one vote and that only the shareholders and proxy holders of shareholders can participate to the voting:

- the proposed resolutions concerning the amendment of the rules on written decision-making in the Board of Directors, the authorisation given to the Board of Directors to pay interim dividends and the revision of the articles of association to bring them into line with Belgian regulations, can be validly adopted if the shareholders attending the meeting, in person or by proxy, represent at least half of the capital, subject to the approval by at least 75% of the votes, in accordance with article 37.3 of the articles of association of the Company *juncto* article 7:153 of the Belgian Code of Companies and Associations; and
- the other proposed resolutions can be validly adopted irrespective of the capital represented by the shareholders attending the meeting in person or by proxy, subject to the approval by at least the majority of the votes cast, in accordance with article 37.3 of the articles of association of the Company.

The chairman remarks that the voting instructions of the shareholders who have voted by correspondence have already been fed into the database of this electronic system and that these will be automatically added to the votes cast at the meeting. The exact totals of the votes by correspondence and the votes cast at the meeting will be enacted in the minutes.

QUESTIONS

The chairman recalls that, as mentioned above, the shareholders could only make use of the possibility to ask their questions in writing.

Questions are answered orally during the meeting.

The written questions and the answers given by the board of directors of the Company remain attached to the present deed.

DETERMINATION OF THE VALIDITY OF THE MEETING

This statement is verified and found to be correct by the meeting, which acknowledges that it is competent to deliberate on the items on the agenda.

The shareholders declare that the shares with which they participate in this extraordinary shareholders' meeting are not subject to any pledge or any other restriction which would prevent the free exercise of their voting rights.

DELIBERATION - RESOLUTIONS

The chairman then submits each of the proposed resolutions on the agenda to the voting of the shareholders.

EXTRAORDINARY SHAREHOLDERS' MEETING

REVISED TEXT OF ARTICLES OF ASSOCIATION

FIRST RESOLUTION: Amendment to the rules of written decision-making in the Board of Directors.

The chairman submits to the meeting the proposal to allow the Board of Directors to decide by way of written resolutions under the conditions of the Belgian Code of Companies and Associations, as follows.

Consequently, the chairman submits to the meeting the proposal to modify <u>article 24.4</u> of the articles of association of the Company as follows:

"24.4 Any or all of the directors may participate in a meeting of the Board of Directors by means of telephone, videoconference or similar communications equipment by means of which all persons participating in the meeting can hear each other. Participation in a meeting by such means shall constitute presence in person at such meeting. Decisions of the Board of Directors may also be adopted, without any physical meeting, by the unanimous consent of the directors expressed in writing."

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,520,110,516
AGAINST	830,697
ABSTENTION	640,486

The resolution was consequently approved.

SECOND RESOLUTION: Power granted to the Board of Directors to distribute interim dividends.

The chairman submits to the meeting the proposal to allow the Board of Directors to distribute interim dividends under the conditions of the Belgian Code of Companies and Associations.

Consequently, the chairman submits to the meeting the proposal to modify <u>article 44</u> of the articles of association of the Company as follows:

"The annual dividends shall be paid at the dates and places decided by the Board of Directors.

The Board of Directors may pay an interim dividend in accordance with article 7:213 of the Code."

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,520,053,764
AGAINST	850,488
ABSTENTION	677,447

The resolution was consequently approved.

THIRD RESOLUTION: Amendment of the articles of association to conform them with Belgian regulation.

The chairman submits to the meeting the proposal to amend several provisions of the articles of association in order to conform such provisions with the changes imposed by or resulting from the rules of the Belgian Code of Companies and Associations and other Belgian

legislations and to align the text of the articles of association to the terminology and numbering of such Code.

Consequently, the chairman submits to the meeting the proposal to adopt the revised text of the articles of association.

The revised text of the articles of association reads as follows:

"Article 1. NAME

The company is a public limited liability company (société anonyme / naamloze vennootschap) with the name "Anheuser-Busch InBev", in short "AB InBev" (the Company). It is subject to the rules applicable to listed public limited liability companies under the Belgian Code of Companies and Associations (the Code).

Article 2. REGISTERED OFFICE - WEBSITE

The registered office is situated at 1 Grand' Place, 1000 Brussels.

The board of directors of the Company (the Board of Directors) may by resolution transfer the registered office to any other town or municipality in Belgium.

The Company may by resolution of the Board of Directors establish seats of administration or operation, branch offices, offices and agencies both in and outside Belgium.

The Company's website can be accessed at ab-inbev.com.

Article 3. DURATION

The Company is incorporated for an unlimited duration.

It may be wound up by resolution of the shareholders' meeting of the Company (the Shareholders' Meeting) passed in the conditions and forms required for a modification of these articles of association (the Articles of Association).

Article 4. CORPORATE PURPOSE

The Company's corporate purpose is:

- a) to produce and deal in all kinds of products, including (but not limited to) beers, drinks, foodstuffs and any ancillary products, as well as all by-products and accessories, of whatsoever use, origin, purpose or form, and to provide all kinds of services; and
- b) to acquire, hold and manage direct or indirect shareholdings or interests in companies, undertakings or other entities having a corporate purpose similar or related to, or likely to promote directly or indirectly the attainment of the foregoing corporate purpose, in Belgium and abroad, and to finance such companies, undertakings or other entities by means of loans, quarantees or in any other manner whatsoever.

In general, the Company may engage in any commercial, industrial and financial transactions, in moveable and real estate transactions, in research and development projects, as well as in any other transaction likely to promote directly or indirectly the attainment of its corporate purpose.

Article 5. CAPITAL AND CLASSES OF SHARES

- 5.1 The share capital of the Company amounts to EUR 1,238,608,344.12. It is represented by 2,019,241,973 shares without nominal value, each of which represents an equal portion of the share capital (the Shares). The share capital has been fully and unconditionally subscribed for and is fully paid up.
- 5.2 There are two classes of Shares. All Shares are ordinary shares (the Ordinary Shares), except for 325,999,817 Shares (the Restricted Shares). The Restricted Shares shall always be in registered form and shall not be listed or admitted to trading on any regulated or unregulated stock market. The holders of Restricted Shares shall be referred to together as the Restricted Shareholders.

5.3 All Shares entitle their holders to the same rights and benefits, except as set out in these Articles of Association.

Article 6. FORM OF THE SHARES AND OTHER SECURITIES

- 6.1 Shares that are not fully paid up shall be in registered form. Shares that are fully paid up and other securities issued by the Company shall be either in registered or dematerialised form, except that Restricted Shares shall always remain in registered form.
- 6.2 Except in the cases referred to in Article 6.1, security holders may elect to have, at any time, at their own expense, registered Shares or other registered securities converted into dematerialised Shares or other dematerialised securities (and vice versa). Dematerialised Shares or other dematerialised securities are represented by a book-entry in an account opened in the name of their owner or holder with an authorised account holder or with a clearing institution.
- 6.3 The register of registered Shares and the register of other registered securities issued by the Company, if any, are held in electronic form by the Company. The Board of Directors may decide to outsource the maintenance and administration of any electronic register to a third party. All entries in the registers, including transfers and conversions, can validly be made on the basis of documents or instructions which the transferor, transferee and/or holder of the securities, as applicable, may send electronically or by other means. The Company may oppose the registration in the register of registered Shares of any transfers, pledges or conversions that do not comply with the terms of these Articles of Association.
- 6.4 The Board of Directors may decide to split the register of registered Shares into two volumes, one to be kept at the registered office of the Company and one to be kept elsewhere, in accordance with the conditions set out in article 7:33 of the Code.

Article 7. TRANSFER OF SHARES – PLEDGES - CONVERSION

- 7.1 Ordinary Shares are freely transferable.
- 7.2 Subject to Article 7.3, no Restricted Shareholder shall transfer, sell, contribute, offer, grant any option on, otherwise dispose of, pledge, charge, assign, mortgage, grant any lien or any security interest on, enter into any certification (certification / certificering) or depository arrangement or enter into any form of hedging arrangement with respect to, in each case directly or indirectly, any of its Restricted Shares or any interests therein or any rights relating thereto, or enter into any contract or other agreement to do any of the foregoing, for a period of five years expiring on 10 October 2021.
 - 7.3 Notwithstanding Article 7.2:
- (a) any Restricted Shareholder may transfer, sell, contribute, offer, grant any option on, otherwise dispose of, pledge, charge, assign, mortgage, grant a lien or any security interest on, or enter into any form of hedging arrangement with respect to, in each case directly or indirectly, any of its Restricted Shares or any interests therein or any rights relating thereto, or enter into any contract or other agreement to do any of the foregoing, to or for the benefit of any person that is its Affiliate, its Successor and/or Successor's Affiliate (together a Restricted Shareholder Group), provided that if any such transferee ceases to be a member of the Restricted Shareholder Group of the Restricted Shareholder that initially made the transfer (or of its Successor), all such Restricted Shares which such transferee owns or in which it holds an interest shall be automatically transferred to such Restricted Shareholder (or to a person which, at the time of such transfer, is its Affiliate or its Successor) and shall therefore remain Restricted Shares;

for the purposes of these Articles of Association, an Affiliate of any person has the meaning given to it in article 1:20 of the Code, and a Successor of any person shall mean (i) in respect of any entity, any entity (x) to which such person transfers all of its assets and (y) which is (and continues to be) directly or indirectly controlled solely or jointly (within the meaning of articles 1:14, 1:17 and

- 1:18 of the Code) by the same entities (or their Successors) or individuals (or any heirs of such individuals) that exercised directly or indirectly sole or joint control over, such shareholder immediately prior to such transfer, or (ii) in respect of any individual, any heir of that individual following his or her death or any individual to whom the assets of such individual are required to be transferred by virtue of applicable law;
- (b) any Restricted Shareholder may (or, for the purposes of Article 7.3(b)(ii)(aa) only, a Pledgee or a Receiver as defined below -may):
- (i) with the prior written consent granted by the Board of Directors (a Pledge Consent) (it being understood that the consent letters entered into by Anheuser-Busch InBev SA/NV with Altria Group, Inc. and BEVCO Ltd. on 11 November 2015 (as subsequently amended from time to time) constitute Pledge Consents for all purposes under these Articles of Association), pledge, charge, assign, mortgage, or otherwise grant a lien over or grant any security interest on all or any part of its Restricted Shares or any interests therein and any rights relating thereto as security (in each case, a Pledge) in respect of any bona fide loans, credit facilities, notes, surety bonds (or other arrangements to secure a stay of execution on or the satisfaction of a judgment or order), letters of credit or any similar extensions of credit to such Restricted Shareholder or any of its Affiliates, hedging, derivative or other financing transactions to which such Restricted Shareholder or any of its Affiliates is a party or, in each case, in respect of which such Restricted Shareholder or any of its Affiliates is a guarantor or security provider, or a guaranty of any of the foregoing;
- (ii) transfer, sell, contribute, offer, grant any option on, or otherwise dispose of, in each case directly or indirectly, or enter into any contract or other agreement to do any of the foregoing in respect of all or part of (or any interest in) the Restricted Shares that are the subject of a Pledge to which a Pledge Consent has been given:
- (aa) to, or as directed by or with the written consent of, the relevant pledgee, chargee, assignee, mortgagee, or other security interest holder (a Pledgee) or to, or as directed by or with the written consent of, a receiver, administrator or other similar official appointed in connection with an enforcement of a Pledge (a Receiver), simultaneously with, or at any time after, such Restricted Shareholder, Pledgee or Receiver notifying the Company that such Pledgee or Receiver has enforced or commenced enforcement action with respect to such Pledge; or
- (bb) to the extent the Restricted Shareholder determines in good faith that such transfer is the only commercially reasonable alternative available to prevent an imminent enforcement of a Pledge by a Pledgee or a Receiver in respect of such Restricted Shares (and the proceeds of the transfer are used to satisfy the underlying obligation secured by the Pledge) and has given written notice to the Board of Directors in which the Restricted Shareholder confirms that it has determined in good faith that such transfer is the only commercially reasonable alternative available to prevent an imminent enforcement of a Pledge by the relevant Pledgee or Receiver in respect of such Restricted Shares.

In these Articles of Association, a Restricted Transferee shall mean each of any Pledgee, Receiver, anyone to whom any Restricted Shares (or any interest in those Restricted Shares) are (or are agreed to be) transferred, sold, contributed, offered, granted any option on, or otherwise disposed of in accordance with Article 7.3(b)(ii)(aa) or 7.3(b)(ii)(bb), and any person referred to in Article 7.5(c).

7.4 The Company shall record in the register of registered Shares in accordance with Belgian law the details of any Pledge notified to it and which is permitted in accordance with a Pledge Consent, by the end of the next Business Day (as defined below) following the day on which it has received notice of such Pledge and shall, if requested, provide evidence thereof to the relevant shareholder as soon as practicable after such recordation.

- 7.5 A Restricted Share shall be unconditionally convertible at the option of the person or persons specified below into Ordinary Shares (on the basis of one Ordinary Share for each Restricted Share):
- (a) at any time after 10 October 2021, at the option of the holder in respect of all or any portion of its Restricted Shares;
- (b) immediately prior to, but then solely for the purpose of facilitating, or at any time after entering into an agreement or arrangement to effect, any transfer, sale, contribution, offer, or other disposal permitted pursuant to Article 7.3(b)(ii), at the option of the holder of the Restricted Shares which are the subject of such transaction or of the Restricted Transferee in respect of such Restricted Shares; and/or
- (c) at the option of (i) a Pledgee under a Pledge in respect of shares of SABMiller plc in respect of which a Pledge Consent has been granted or a Receiver in respect of such shares of SABMiller plc where the Pledgee has exercised its rights of enforcement (itself or through a Receiver) prior to 10 October 2016 or (ii) a transferee of, in lieu of, or at the direction of, any such Pledgee or Receiver, in respect of all or any portion of its Restricted Shares.
- 7.6 Any Restricted Share shall automatically be converted into an Ordinary Share (on the basis of one Ordinary Share for each Restricted Share):
- (a) upon any transfer, sale, contribution or other disposal of any such Restricted Share or any interests or rights therein (including to a Restricted Transferee), except only in the instances referred to in Articles 7.3(a) and 7.3(b)(i) provided that, in such instances, the Restricted Shares shall automatically be converted into Ordinary Shares upon any subsequent transfer, sale, contribution or disposal to any party which is not an Affiliate, a Successor or a Successor's Affiliate of the Restricted Shareholder; in no case whatsoever shall a Restricted Transferee ever become a Restricted Shareholder in respect of such Restricted Shares;
- (b) immediately prior to the closing of a successful public takeover bid for all the Shares in the Company or the completion of a merger of the Company as acquiring or disappearing company, in circumstances where the shareholders directly or indirectly controlling (within the meaning of article 1:14 of the Code) or exercising directly or indirectly joint control (within the meaning of article 1:18 of the Code) over the Company immediately prior to such takeover bid or merger will not directly or indirectly control, or exercise directly or indirectly joint control over, the Company or the surviving entity following such takeover bid or merger; and/or
- (c) upon the announcement of a squeeze-out bid for the outstanding Shares, in accordance with article 7:82 of the Code.
- 7.7 In the event that all the Shares in the Company are acquired by a company which the shareholders of the Company immediately prior to such acquisition directly or indirectly control or exercise directly or indirectly joint control over (within the meaning of articles 1:14, 1:17 and 1:18 of the Code), Restricted Shareholders shall be treated in an equivalent manner to holders of Ordinary Shares, save that there shall be equivalent differences between the rights and restrictions attaching to the shares to be issued to holders of Ordinary Shares and the shares to be issued to holders of Restricted Shares to reflect the differences in rights and restrictions between the Ordinary Shares and the Restricted Shares.
- 7.8 Save where a Pledge Consent has previously been given to a Restricted Shareholder, in which event no further authorisation or consent of, or action by, the Board of Directors shall be required with respect to the creation, existence or enforcement of, or the exercise of the rights under, any Pledge made by such Restricted Shareholder or any of its Affiliates in accordance with the terms of such Pledge Consent, the Board of Directors shall have absolute discretion as to whether to grant a Pledge Consent. The pledging policy that the Board of Directors shall adopt from

time to time will set out the circumstances in which the Board of Directors will grant a Pledge Consent. The provisions of these Articles of Association shall be prevailing in the event of any conflict between the pledging policy and these Articles of Association.

7.9 Provided that the relevant conditions set forth in Article 7.5 or 7.6, as applicable, are met, the Company will record the conversion of the Restricted Shares into Ordinary Shares (and, if relevant, the transfer of the Ordinary Shares which are the subject of the conversion of the Restricted Shares to or as directed in writing by the relevant Restricted Shareholder or the Restricted Transferee), in the Company's share register, on the same Business Day (if the notice relating to such conversion and/or transfer is received by the Company before 1:00 pm Belgian time) or the next Business Day (if the notice relating to this conversion and/or transfer is received by the Company after 1:00 pm Belgian time or on a day that is not a Business Day). For the purpose of these Articles of Association, Business Day(s) shall mean all days of the week with the exception of Saturdays, Sundays and legal public holidays in Belgium.

From the time of conversion, the Ordinary Shares will be freely and unconditionally transferable by (or as directed by) the Restricted Shareholder or Restricted Transferee or any transferee or transferees of such Ordinary Shares, free of any lock-up or other restriction.

On the same day as the recordation of the conversion, the Company will send to Euronext Brussels a request for admission to listing of such Ordinary Shares and take all such steps that are within the Company's control to ensure that admission to listing occurs promptly thereafter. Neither the Restricted Shareholder, the Restricted Transferee nor any other transferee shall be liable for any costs or expenses incurred by the Company in connection with such conversion or transfer, and the Company shall not be liable for any delay in conversion or transfer or admission to listing provided it complies with this Article 7.9.

7.10 Upon conversion of Restricted Shares in accordance with this Article 7, the Board of Directors shall be empowered to amend Article 5.2 to reflect the modification in the number of Ordinary Shares and Restricted Shares.

Article 8. RIGHTS OF RESTRICTED SHARES

- 8.1 If at any time the Ordinary Shares shall be changed into a different number of Shares or a different class of Shares by reason of any share dividend, subdivision, reorganisation, reclassification, recapitalisation, stock split, reverse stock split, combination or exchange of Shares, or any similar event shall have occurred, there will be an equivalent share dividend, subdivision, reorganisation, reclassification, recapitalisation, stock split, reverse stock split, combination or exchange of Shares or similar event with respect to the Restricted Shares, provided that (i) nothing shall be deemed to permit the Company (including the Board of Directors) to take any action with respect to its share capital that is otherwise prohibited by these Articles of Association and (ii) if any such event would otherwise cause any Restricted Shareholder to cease to hold at least one such Restricted Share by virtue of its entitlement following such event being to a fraction of less than one Restricted Share, its entitlement following such event shall be rounded up to one Restricted Share. In case of any event referred to in this Article 8.1, Restricted Shareholders shall only be entitled or required to receive Restricted Shares in respect of the Restricted Shares held by them.
- 8.2 As long as there remain any Restricted Shares, any modification of the rights attached to the Ordinary Shares or the Restricted Shares shall be made in accordance with the quorum and majority requirements of article 7:155 of the Code.

Article 9. ORDERLY DISPOSAL

Any initial holder of Ordinary Shares resulting from the conversion of Restricted Shares which were previously held by such holder (other than a Restricted Transferee) or any of its Affiliates who contemplates selling such Ordinary Shares on a stock exchange on which the Ordinary Shares

are listed (other than by a block trade or overnight placement in accordance with customary market practice for dispositions of such nature) in a single transaction or series of connected transactions for an amount exceeding 1% of the total share capital within three months of the date of conversion shall use reasonable endeavours to effect such sale in an orderly manner of disposition that is not likely to disrupt materially the market for the Shares and shall consult with the Company in advance of such sale, subject to the Company consenting to being made an insider for these purposes. For the avoidance of doubt, this Article 9 shall not apply to transfers to any Restricted Transferee in the circumstances set out in Article 7.3(b)(ii).

Article 10. DISCLOSURE OF SIGNIFICANT SHAREHOLDINGS

In addition to the transparency disclosure thresholds of 5% and multiples of 5% set out by the applicable Belgian legislation, the disclosure obligation set out in such legislation shall also apply with respect to the thresholds of 3% and 7.5%.

Article 11. AUTHORISED CAPITAL

11.1 The Board of Directors may increase the share capital of the Company, in one or several times, by the issuance of a number of Shares, or financial instruments giving right to Shares, which will not represent more than 3% of the Shares outstanding as at 26 April 2017 (the amount obtained shall be, to the extent necessary, rounded down to result in an entire number of Shares), provided that, in accordance with article 7:198, first indent, of the Code, this may not result in the share capital being increased, in one or several times, by an amount exceeding the amount of share capital prevailing on 26 April 2017.

The increase(s) of capital decided under such authorisation may be effected by contribution in cash or in kind, including as the case may be an issue premium not available for distribution, the amount of which shall be fixed by the Board of Directors, and by creation of new Shares conferring such rights as the Board of Directors shall determine.

The increase(s) of capital may also be effected by capitalisation of reserves, including those not available for distribution, or an issue premium, with or without the creation of new Shares.

Such authorisation is granted to the Board of Directors for a period of five years as from the date of publication of the amendment of the Articles of Association resolved upon by the extraordinary Shareholders' Meeting of 26 April 2017. It can be renewed, once or several times, in accordance with applicable legal provisions.

11.2 The Board of Directors may be expressly authorised by the Shareholders' Meeting, in case of public take-over bids on securities of the Company, to increase the capital, under the conditions set out in article 7:202 of the Code. This authorisation may be granted for a period of three years as from the date of the amendment of the Articles of Association setting out the authorisation.

Article 12. INCREASE AND DECREASE OF CAPITAL – PREFERENTIAL SUBSCRIPTION RIGHT

12.1 In case of an increase of capital through the issue of new Shares, convertible bonds, bonds repayable in Shares, subscription rights or other financial instruments giving a right to Shares (any such Shares, bonds, rights or instruments being referred to as Equity Interests), all shareholders will have a preferential right to subscribe for any such Equity Interests, as set out in and in accordance with article 7:188 of the Code. The preferential subscription right shall entitle each shareholder to subscribe for any new Equity Interests, pro rata to the proportion of existing share capital as it holds immediately prior to such issue and subject to the rules of article 7:188 of the Code. Each shareholder may exercise its preferential right in whole or in part.

The time within which the preferential subscription right may be exercised shall be fixed by the Shareholders' Meeting or by the Board of Directors as the case may be, but shall not be less

than fifteen calendar days from the date on which the subscription period was opened. The preferential subscription right shall be negotiable during the subscription period to the extent to which the Shares may be transferred.

Subject to the rules of article 7:188 of the Code, the Board of Directors may decide that preferential subscription rights which were not or were only partly exercised by any shareholders shall accrue proportionally to the other shareholders of the same class who have already exercised their subscription rights prior to the exercise by the shareholders of the other class of their second ranking preferential subscription right as the case may be. The Board of Directors shall fix the practical terms for such subscription. The Board of Directors may also conclude, upon such terms as it shall determine, all agreements intended to secure the subscription of part or all of the new Shares to be issued.

12.2 The Shareholders' Meeting acting in accordance with article 7:191 of the Code may restrict or cancel the preferential subscription right for a purpose that is in the best interest of the Company, provided however that if the preferential subscription right is restricted or cancelled with respect to any issuance in which any existing shareholder subscribes to any Equity Interests, all existing shareholders shall be given the same right and be treated in the same way. Such equal treatment requirement shall not apply when the preferential subscription right is restricted or cancelled with respect to issuances of Equity Interests issued solely pursuant to any stock option plans or other compensation plans in the ordinary course of business.

In the case the Shareholders' Meeting has granted an authorisation to the Board of Directors to effect a capital increase in the framework of the authorised capital and such authorisation allows the Board of Directors to do so, the Board of Directors may likewise restrict or cancel the preferential subscription right applying the same principles as set out in this paragraph.

No Restricted Shares shall be issued other than to a Restricted Shareholder exercising its preferential subscription right in respect of its holding of Restricted Shares, without prejudice to the right of the Ordinary Shareholders to exercise their second ranking preferential subscription right in accordance with article 7:188 of the Code.

- 12.3 The Company may proceed with a reduction in capital in accordance with articles 7:208 and following of the Code. No such reduction may affect the Restricted Shareholders unless it is approved in accordance with the procedure provided by article 7:155 of the Code, if such article is applicable.
- 12.4 Any issue premium booked by the Company shall be recorded in an account that is not available for distribution and may be reduced or cancelled only by a resolution of the Shareholders' Meeting approved in accordance with the conditions of article 7:208 of the Code.

Article 13. BONDS, SUBSCRIPTION RIGHTS AND OTHER SECURITIES GIVING RIGHT TO SHARES

- 13.1 The Company may issue bonds by resolution of the Board of Directors and on such conditions as it shall determine. The Shareholders' Meeting, or the Board of Directors acting within the framework of the authorised capital, may decide to issue convertible bonds, bonds repayable into Shares, subscription rights or any other financial instrument giving right to Shares, subject to Article 12.
- 13.2 The holders of convertible bonds or subscription rights have the right to attend the Shareholders' Meetings, but only in a consultative capacity.

Article 14. PAYMENTS

14.1 The Board of Directors may make calls upon the shareholders in such amounts and at such times as it shall fix in respect of any moneys to be paid on the Shares issued further to a capital increase.

Any shareholder who, after fifteen calendar days as from notice given by registered letter, remains in default of payment, shall pay the Company interest at the statutory rate plus two per cent as from the due date. Where such failure is not remedied within one month of a second notice, the Board of Directors may declare the shareholder's rights forfeited and cause the Shares to be sold without prejudice to the right to claim from it any remainder outstanding plus such damages as may apply.

14.2 The Board of Directors may authorise shareholders to pay in anticipation the moneys uncalled on their Shares on such terms as it shall fix.

Article 15. ACQUISITION BY THE COMPANY OF ITS OWN SHARES

- 15.1 The Company may, without any prior authorisation of the Shareholders' Meeting, in accordance with article 7:215 of the Code where applicable, acquire, on or outside the stock exchange, its own Shares up to a maximum of 20% of the outstanding Shares of the Company at the moment of the acquisition for a unitary price which will not be lower than one euro (EUR 1,-) and not higher than 20% above the highest closing price on Euronext Brussels during the last twenty trading days preceding the acquisition.
- 15.2 The Company may, without any prior authorisation of the Shareholders' Meeting, in accordance with article 7:218 of the Code, dispose, on or outside the stock exchange, of the Shares of the Company which were acquired by the Company under the conditions determined by the Board of Directors.

With respect to the Shares acquired by the Company as a result of the merger between the Company and Anheuser-Busch InBev SA/NV, the Board of Directors shall be entitled to dispose of such Shares only in connection with (i) any Share delivery obligations undertaken by Anheuser-Busch InBev SA/NV prior to 11 November 2015, (ii) any stock option plans or other compensation plans (including the Zenzele scheme), or (iii) any stock lending agreement or similar arrangement in respect of which the Company used the Shares for the purposes set out in items (i) and (ii).

- 15.3 The authorisations set forth in Articles 15.1 and 15.2 also extend to acquisitions and disposals of Shares of the Company by direct subsidiaries of the Company made in accordance with article 7:221 of the Code. Such authorisations were granted for a period of five years as from 28 September 2016.
- 15.4 Indirect subsidiaries of the Company may, without any prior authorisation of the Shareholders' Meeting, acquire or dispose of the Shares without any limitation in time, under the conditions set out in article 7:222 of the Code.

Article 16. INDIVISIBILITY OF SECURITIES

All securities shall be held in undivided ownership vis-à-vis the Company. Without prejudice to Article 34 relating to representation at the Shareholders' Meeting, the Company may suspend all rights attaching to securities until such time as one person shall have been appointed holder thereof vis à vis the Company.

Article 17. SUCCESSORS IN TITLE

Subject to the other provisions of these Articles of Association, the rights and obligations attaching to a Share follow that Share, regardless of whom it is transferred to.

Neither the heirs nor the creditors of a shareholder may, on any grounds whatsoever, require the division or sale by auction of the Company's assets, nor interfere in any way whatsoever with the administration of the Company. They shall, in exercising their rights, abide by the annual accounts and decisions of the Shareholders' Meeting.

Article 18. CERTIFICATION OF THE SECURITIES OF THE COMPANY

18.1 The Board of Directors may resolve that the Company will give assistance to a third party for the issuance by the third party of certificates, under the conditions set out by law, in order

to represent securities issued by the Company. It may resolve that the Company will pay all or part of the charges of such certification and of the setting up and operating charges of the issuer of the certificates, insofar as such payment is in the interest of the Company.

A certificate holder or issuer or any third party of any kind may only invoke the assistance of the Company in their issuing if the Company has confirmed this assistance in writing to the issuer. The holders of such certificates may only exercise rights towards the Company that are granted to them by law if the evidence of ownership of the registered certificates have previously been approved in writing by the Company.

18.2 Any issuer of certificates, whether or not issued with the assistance of the Company, intending to participate in a Shareholders' Meeting and exercise the voting rights attached to the certified securities, shall comply with the formalities described in Articles 33.1 to 33.3. Any holder of certificates issued with the assistance of the Company, intending to attend a Shareholders' Meeting in a consultative capacity, shall comply with the formalities described in Article 33.4.

Article 19. COMPOSITION OF THE BOARD OF DIRECTORS

- 19.1 The Company shall be managed by a Board of Directors comprising a minimum of three and a maximum of fifteen directors. The directors may be natural persons or legal entities and may but need not be shareholders. The directors are appointed by the Shareholders' Meeting and are dismissible by it at any time.
- 19.2 When a legal entity is appointed as a director, it must specifically appoint a natural person as its permanent representative, to carry out the office of director in the name and on behalf of the legal entity. The legal entity may not revoke its permanent representative without simultaneously appointing a successor. The appointment and termination of the office of the permanent representative are governed by the same disclosure rules as if he/she were exercising the office on his/her own behalf.
 - 19.3 The Board of Directors shall be composed as follows:
- (a) three directors shall be independent directors appointed by the Shareholders' Meeting upon proposal by the Board of Directors;
- (b) so long as the Stichting Anheuser-Busch InBev and/or any of its Affiliates, any of their respective Successors or Successors' Affiliates (together, the Reference Shareholder) own, in aggregate, more than 30% of the Shares with voting rights in the share capital of the Company, nine directors shall be appointed by the Shareholders' Meeting upon proposal by the Reference Shareholder; and
- (c) so long as the Restricted Shareholders together with their Affiliates, any of their respective Successors and/or Successors' Affiliates own, in aggregate (and taking into account the Ordinary Shares referred to in Article 20.2(b)):
- (i) more than 13.5% of the Shares with voting rights in the share capital of the Company, three directors shall be appointed by the Shareholders' Meeting upon proposal by the Restricted Shareholders in accordance with the procedure set out in Article 21 (each director appointed in accordance with such procedure being a Restricted Share Director);
- (ii) more than 9% but not more than 13.5% of the Shares with voting rights in the share capital of the Company, two Restricted Share Directors shall be appointed;
- (iii) more than 4.5% but not more than 9% of the Shares with voting rights in the share capital of the Company, one Restricted Share Director shall be appointed; and
- (iv) 4.5% or less than 4.5% of the Shares with voting rights in the share capital of the Company, they shall no longer have the right to propose any candidate for appointment as a member of the Board of Directors and no Restricted Share Director shall be appointed;

it being understood that, for the purpose of determining the number of directors to be appointed upon proposal of the Reference Shareholder and the Restricted Shareholders, the percentage of Shares with voting rights held respectively by the Reference Shareholder and the Restricted Shareholders (together with their Affiliates, respective Successors and/or Successors' Affiliates) shall be computed in accordance with the rules set out in Article 20.

- 19.4 The term of office of directors shall be as follows:
- (a) with respect to all directors except the Restricted Share Directors, unless the Shareholders' Meeting sets a shorter term, the term of office shall terminate immediately after the closing of the fourth ordinary Shareholders' Meeting following the date of their appointment (or such shorter term decided by the Shareholders' Meeting);
- (b) with respect to all Restricted Share Directors, the term of office shall terminate immediately after the closing of the next ordinary Shareholders' Meeting following the date of their appointment; and
 - (c) all directors shall be eligible for re-election.
- 19.5 If, any time between two ordinary Shareholders' Meetings, the number of Restricted Share Directors which the Restricted Shareholders are entitled to propose for appointment to the Board of Directors pursuant to Articles 19.3(c), 20.2(a) and 20.2(b) (the Permitted Number) becomes and remains lower than the number of Restricted Share Directors that are members of the Board of Directors, the Board of Directors shall remain composed of the same number of Restricted Share Directors until the next ordinary Shareholders' Meeting. At such Shareholders' Meeting, the Restricted Shareholders shall be entitled to present only a number of candidates for appointment to the Board of Directors equal to the Permitted Number.

Article 20. CALCULATION OF THE NUMBER OF DIRECTORS TO BE PRESENTED BY THE REFERENCE SHAREHOLDER AND THE RESTRICTED SHAREHOLDERS

- 20.1 For the purpose of calculating the percentage of Shares with voting rights in the share capital of the Company owned by the Reference Shareholder in order to determine the number of directors to be proposed by the Reference Shareholder under Article 19.3, the following rules shall apply:
- (a) any Shares in the Company (i) issued pursuant to stock option plans or other compensation plans after 10 October 2016, (ii) disposed of by the Company pursuant to Article 15.2, to the extent such shares were owned by the Company as at 10 October 2016, or (iii) owned by the Company or any of its subsidiaries within the meaning of article 1:15 of the Code on the date set out in Article 20.1(b), shall be disregarded for the purposes of calculating the total number of shares with voting rights in the share capital of the Company; and
- (b) the percentage of Shares owned by the Reference Shareholder shall be calculated on the basis of the number of Shares owned by it on the 120th calendar day prior to the relevant Shareholders Meeting at which the appointment, re-election or confirmation of co-optation of any director proposed by the Reference Shareholder is to be decided.
- 20.2 For the purpose of calculating the percentages of Shares with voting rights in the share capital of the Company owned by a Restricted Shareholders Group (or all Restricted Shareholder Groups in aggregate) in order to determine the number of Restricted Share Directors to be proposed under Article 19.3, the following rules shall apply:
- (a) any Shares in the Company (i) issued pursuant to stock option plans or other compensation plans after 10 October 2016, (ii) disposed of by the Company pursuant to Article 15.2, to the extent such shares were owned by the Company as at 10 October 2016, or (iii) owned by the Company or any of its subsidiaries within the meaning of article 1:15 of the Code on the date set out

in Article 20.2(c), shall be disregarded for the purposes of calculating the total number of shares with voting rights in the share capital of the Company;

- (b) with respect to each person who, as at 10 October 2016, owned Restricted Shares in its own name, as long as such person or any member of its Restricted Shareholder Group still owns at least one Restricted Share in its own name, any Ordinary Shares owned by, or on behalf of, such Restricted Shareholder Group shall be added to the number of remaining Restricted Shares owned by such Restricted Shareholder Group provided that, on the date set out in Article 20.2(c), such Ordinary Shares are in registered form:
- (i) in the name of any member of such Restricted Shareholder Group which member owns at least one Restricted Share;
- (ii) in the name of any member of such Restricted Shareholder Group which member does not itself own at least one Restricted Share and, where at the latest on the tenth calendar day following the date set out in Article 20.2(c), the Company has received (aa) a notice from the member of the Restricted Shareholder Group owning such Ordinary Shares confirming that it is a member of a Restricted Shareholder Group and identifying which such Restricted Shareholder Group, and (bb) a notice from one or more members of the relevant Restricted Shareholder Group which own(s) at least one Restricted Share confirming that such person is a member of the Restricted Shareholder Group;
- in the name of a custodian owning such Ordinary Shares on behalf of any member (iii) of such Restricted Shareholder Group which has the right to exercise the voting rights in respect of such Ordinary Shares as a result of holding an interest in such Ordinary Shares, and where, at the latest on the tenth calendar day following the date set out in Article 20.2(c), the Company has received (aa) a notice from the relevant custodian confirming that, on the date set out in Article 20.2(c), such Ordinary Shares were owned by such custodian on behalf of such member of such Restricted Shareholder Group, and (bb) a notice from such member of such Restricted Shareholder Group confirming that, on the same date, it is a member of the Restricted Shareholder Group owning at least one Restricted Share and identifying which such Restricted Shareholder Group and confirming that such Ordinary Shares were owned on its behalf by such custodian (it being understood that Ordinary Shares will not be deemed to be owned on behalf of a Restricted Shareholder Group by a custodian as a result only of (x) being loaned to such custodian or being subject to any similar arrangement, or (y) being owned by a custodian in connection with a derivative or hedging arrangement between the custodian and the relevant member of the Restricted Shareholder Group except where the custodian directly or indirectly acquired or received such Ordinary Shares from a member of the Restricted Shareholder Group);
- (c) the number and percentage of Restricted Shares and Ordinary Shares held by a Restricted Shareholder Group in accordance with Articles 19.3(c), 20.2(a), 20.2(b) and 21.3 shall be calculated on the basis of the number of such Shares held by it on the 120th calendar day prior to the relevant Shareholders' Meeting at which the appointment, re-election or confirmation of cooptation of a Restricted Share Director is to be decided (each such Shareholders' Meeting being an Appointment Shareholders' Meeting), except as follows:
- (i) if the Appointment Shareholders' Meeting is not the ordinary Shareholders' Meeting referred to in Article 32.1, where this number and percentage shall be calculated on such other date determined by the Board of Directors and announced not later than the tenth calendar day prior to such date; or
- (ii) in the instance set out in Article 22.1(c), where this number and percentage shall be calculated on the date set out in Article 22.4;

- (d) if any person which is a member of a Restricted Shareholder Group ceases to be part of the Restricted Shareholder Group referred to in the notice provided in accordance with Article 20.2(b)(ii) or 20.2(b)(iii), it shall promptly notify the Company; and
- (e) in any event, the Restricted Shareholders (together with any other person which is a member of any Restricted Shareholder Group) shall never be entitled to have more than three directors appointed upon their proposal.

Article 21. APPOINTMENT OF RESTRICTED SHARE DIRECTORS

- 21.1 Except where:
- (a) the Board of Directors has received from a Requisite Majority (as defined below) written resolutions in compliance with the rules set out in Articles 21.4 to 21.6 proposing a sufficient number of candidates so that, if such candidates are appointed or re-elected or their co-optation is confirmed, the Board of Directors will be composed of the Permitted Number of Restricted Share Directors, or
- (b) the confirmation of the co-optation of any Restricted Share Directors is to be submitted to a Shareholders' Meeting which is not otherwise an Appointment Shareholders Meeting and the fact that such confirmation is to be submitted to such Shareholders' Meeting is announced less than 130 calendar days prior to the date of the Shareholders' Meeting,

the Board of Directors shall convene a meeting of the Restricted Shareholders (a Restricted Shareholders' Meeting) at least 90 calendar days in advance of any Appointment Shareholders' Meeting in order for Restricted Shareholders to vote for the candidates to be presented for reelection, appointment or confirmation of co-optation as Restricted Share Director, at such Appointment Shareholders' Meeting. Such Restricted Shareholders' Meeting shall be held at least 60 calendar days prior to the Appointment Shareholders' Meeting.

- 21.2 At least 10 calendar days prior to any Restricted Shareholders' Meeting, any Restricted Shareholder may propose one or more candidates to be presented for appointment as Restricted Share Directors to the Board of Directors and the other Restricted Shareholders. During the Restricted Shareholders' Meeting, Restricted Shareholders shall vote to select the Restricted Share Directors to be appointed among such candidates in one round of voting in accordance with the rules set out in Article 21.3.
- 21.3 The following rules shall apply between the Restricted Shareholders in respect of any Restricted Shareholders' Meeting to select candidates to be presented to the Board of Directors for re-election, appointment or confirmation of co-optation as Restricted Share Directors:
- (a) each Restricted Shareholder Group shall have one vote for each Restricted Share and each Ordinary Share referred to under Article 20.2(b) (all such Shares together the Restricted Shareholder Voting Shares), provided that:
- (i) if the aggregate of the Restricted Shareholder Voting Shares owned by, or on behalf of, any Restricted Shareholder Group exceeds 175% of the aggregate number of Restricted Shares held by such Restricted Shareholder Group on 10 October 2016 (or the relevant adjusted number taking account of any change to Restricted Shares as referred to in Article 8.1), such Restricted Shareholder Group shall together not be entitled to exercise more than the number of votes equal to 175% of the number of the Restricted Shares owned by, or on behalf of, such Restricted Shareholder Group on 10 October 2016 (or the relevant adjusted number taking account of any change to Restricted Shares as referred to in Article 8.1);
- (ii) if any Restricted Shareholder Group wishes to exercise the additional votes attaching to any Ordinary Shares referred to in Article 20.2(b) owned by, or on behalf of, such Restricted Shareholder Group, such Restricted Shareholder Group must cast all of the votes attaching to its Restricted Shareholder Voting Shares in favour of no more than two candidates; and

- (iii) no candidate shall be eligible to receive such additional votes from more than one Restricted Shareholder Group (it being understood that, in the event any one candidate receives such additional votes from more than one Restricted Shareholder Group, such candidate shall be deemed to have received the number of additional votes of the Restricted Shareholder Group casting the most additional votes in favour of such candidate);
- (b) all candidates will be proposed in one single round of voting where all votes attaching to all Restricted Shareholder Voting Shares may be cast;
- (c) subject to Article 21.3(a)(ii), any vote may be cast in favour of any candidate and a Restricted Shareholder may cast its votes in any manner it chooses between the candidates (including by exercising all votes in favour of a single candidate or dividing its votes in any proportion it chooses among more than one candidate);
- (d) the candidate(s) having received the highest number of votes up to the number of Restricted Share Directors to be appointed in accordance with Article 19.3 shall be presented to the Appointment Shareholders' Meeting for appointment; and
- (e) following each Restricted Shareholders' Meeting, the Company shall as promptly as reasonably practicable notify the Restricted Shareholders of (i) the identity of the candidates selected by the Restricted Shareholders for appointment, re-election or confirmation of co-optation as Restricted Share Directors (as applicable) by the ensuing Appointment Shareholders' Meeting and (ii) with respect to each candidate so identified, the Restricted Shareholder Group that cast the most votes in favour of such candidate at such Restricted Shareholders' Meeting (the Proposing Holder).
- 21.4 Decisions of the Restricted Shareholders' Meetings can also validly be taken by written resolutions of a Requisite Majority, provided that (i) one or more persons who constitute or may constitute a Requisite Majority and intend to act by way of written resolutions pursuant to this Article (if they do constitute a Requisite Majority) appoint a representative to communicate with the Board of Directors in connection therewith (a Restricted Shareholder Representative) and (ii) between 115 and 135 calendar days prior to the applicable Appointment Shareholders' Meeting (or in the case of written resolutions in connection with a vacancy pursuant to Article 22.1(c), within 10 calendar days of the vacancy arising), such Restricted Shareholder Representative notifies to the Board of Directors the intention of such person(s) to act by way of written resolution pursuant to this Article 21.4 together with the identity of each such person (a Restricted Shareholder Representative's Notice).

For the purpose of these Articles of Association, a Requisite Majority shall mean persons having the right to exercise the voting rights as a result of holding directly or indirectly not less than the minimum number of Restricted Shareholder Voting Shares that would be necessary to select the Permitted Number of candidates for appointment, re-election or confirmation of co-optation as Restricted Share Directors at a Restricted Shareholders' Meeting at which all Restricted Shareholder Voting Shares were present and voted (subject to the restrictions set out in Article 21.3(a)), irrespective of how any Restricted Shareholder Voting Shares held by other persons would have been voted at such meeting (other than as a result of the restrictions set out in Article 21.3(a)).

21.5 The Company shall, at least 105 calendar days prior to any applicable Appointment Shareholders' Meeting (or in the case of a written resolution in connection with a vacancy pursuant to Article 22.1(c), within 10 calendar days of the Restricted Shareholder Representative's Notice), provide the Restricted Shareholder Representative with (i) a form of written resolutions that may be used by the Requisite Majority to specify candidates for appointment, re-election or confirmation of co-optation as Restricted Share Directors pursuant to Article 21.4 at the next Appointment Shareholders' Meeting or, as applicable, in order to fill a vacancy pursuant to Article 22.1(c), and (ii) a statement of the number of Restricted Share Directors eligible to be appointed (including any re-

elections or confirmations of co-optation) in accordance with Article 19.3(c) or, as applicable, the number of vacancies to be filled in accordance with Article 22.1(c). If the Company concludes that the person(s) identified in the Restricted Shareholder Representative's Notice in relation to any applicable Appointment Shareholders' Meeting do not constitute a Requisite Majority, the Board of Directors shall convene a Restricted Shareholders' Meeting in accordance with Article 21.1. Written resolutions delivered pursuant to Article 21.4 shall specify, with respect to each candidate specified for appointment as a Restricted Share Director by such written resolutions, the Restricted Shareholder Group which shall be deemed to be the Proposing Holder.

- 21.6 The Restricted Shareholder Representative (if any) shall notify the Board of Directors of the identity of the candidate(s) selected to be presented for appointment, re-election or confirmation of co-optation as Restricted Share Directors at the latest 95 calendar days prior to the date of the Appointment Shareholders' Meeting by sending to the Board of Directors a copy of the written resolutions of the Restricted Shareholders (or in the case of written resolutions in connection with a vacancy pursuant to Article 22.1(c), within 30 calendar days following the vacancy arising). The candidates notified to the Board of Directors in accordance with this Article 21.6 shall be presented to the Appointment Shareholders' Meeting for appointment, re-election or confirmation of co-optation, it being understood that the Restricted Shareholders shall not be allowed to notify for appointment, re-election or confirmation of co-optation as Restricted Share Directors in aggregate more candidates than the Permitted Number.
- 21.7 If, for any reason, the Restricted Shareholders propose fewer (or no) candidates for appointment, re-election and confirmation of co-optation as Restricted Share Directors in aggregate at any Appointment Shareholders' Meeting so that, if such candidates are appointed or re-elected or their co-optation is confirmed (as applicable), the Board of Directors will be composed of fewer than the Permitted Number of Restricted Share Directors, the Board of Directors shall be validly composed of a lower number of Restricted Share Directors than is foreseen in Article 19.3 until the next Appointment Shareholders' Meeting.

Article 22. VACANCIES

- 22.1 When a position on the Board of Directors becomes vacant, the remaining directors shall have the right to temporarily fill the vacancy by appointing a candidate proposed by:
 - (a) the Board of Directors, in the case of a vacancy relating to an independent director;
- (b) the Reference Shareholder, in the case of a vacancy relating to a director appointed upon proposal of the Reference Shareholder; or
- (c) in the case of a vacancy relating to a Restricted Share Director, the following in descending order of precedence:
- (i) the applicable Proposing Holder (if such Proposing Holder then holds a Sufficient Restricted Shareholding) by written proposal notified to the Board of Directors at the latest on the 15th calendar day following the vacancy arising;
- (ii) if (i) does not apply, a Requisite Majority of Restricted Shareholders acting by written resolutions notified to the Board of Directors at the latest on the 30th calendar day following the vacancy arising; and
 - (iii) if neither (i) nor (ii) applies, a Restricted Shareholders' Meeting.

Sufficient Restricted Shareholding means, with respect to a Proposing Holder, the direct and indirect holding by such Proposing Holder of no less than the minimum number of Restricted Shareholder Voting Shares that would be necessary for such Proposing Holder to be capable of selecting such number of candidates for appointment to the Board of Directors at a Restricted Shareholders' Meeting at which all Restricted Shareholder Voting Shares were present and voted (subject to the restrictions set out in Article 21.3(a)), irrespective of how any Restricted Shareholder

Voting Shares held by other persons would have been voted at such meeting (other than as a result of the restrictions set out in Article 21.3(a)), as is at least equal in the aggregate to the number of candidates being proposed by such Proposing Holder pursuant to the preceding sentence of this Article 22.1 plus the number of continuing Restricted Share Director (if any) in respect of which it is the Proposing Holder.

- 22.2 Any temporary appointment effected in accordance with Article 22.1 shall (i) be subject to confirmation at the next Shareholders' Meeting unless the Board of Directors, the Reference Shareholder or the Restricted Shareholders' Meeting (or a Requisite Majority of Restricted Shareholders), as applicable, propose an alternative candidate in accordance with the rules set out in Articles 19 to 21, and (ii) subject to such confirmation, be for a term equal to the remainder of the term of the director who held office prior to such vacancy arising.
- 22.3 In the event of a Restricted Shareholders' Meeting in the situation referred to in Article 22.1(c)(iii):
- (a) the Restricted Shareholders' Meeting shall be convened by the Board of Directors at the latest on the 40th calendar day following the vacancy arising and held within 70 calendar days of the vacancy arising;
- (b) at least 10 calendar days prior to any such Restricted Shareholders' Meeting, any Restricted Shareholder may propose one or more candidates to be proposed to the Restricted Shareholders' Meeting as the candidate (in the case of a single vacancy) or candidates (in the case of multiple vacancies) to be proposed for co-optation to the Board of Directors; and
- (c) the candidate (in the case of a single vacancy) or candidates (in the case of multiple vacancies) to be proposed for co-optation to the Board of Directors shall be selected in a single round of voting where all votes attaching to Restricted Shareholder Voting Shares may be cast and the provisions of Article 21.3 shall apply mutatis mutandis thereto.
- 22.4 In the instance set out in Article 22.1(c), the number and percentage of Restricted Shares and Ordinary Shares held by a Restricted Shareholder Group shall be calculated on the basis of the number of such Shares held by it on the 120th calendar day prior to the ordinary Shareholders' Meeting referred to in Article 32.1 held most recently prior to the vacancy arising.

Article 23. CHAIRPERSON OF THE BOARD

The Board of Directors shall elect one of its members to be chairperson, who shall be an independent director appointed in accordance with Article 19.3(a) or a Restricted Share Director appointed in accordance with Article 19.3(c). The Board may elect one or more vice-chairperson(s). The Shareholders' Meeting may, upon proposal of the Board of Directors, confer honorary status on former chairpersons, vice-chairpersons or directors. The Board of Directors may then invite them to attend its meetings in an advisory capacity.

Article 24. BOARD MEETINGS

24.1 The Board of Directors shall meet as frequently as the interests of the Company shall require. The Board of Directors shall hold regular meetings at such times as may be from time to time fixed by a resolution of the Board of Directors.

A meeting of the Board of Directors shall be held without notice immediately before the ordinary Shareholders' Meeting. In addition, special meetings of the Board of Directors may be called and held at any time upon the call of either the chairperson of the Board of Directors or at least two directors, by notice to each director at least three Business Days before the meeting. Reasonable efforts shall be made to ensure that each director actually receives timely notice of any such special meeting. Where duly justified by emergency and by the corporate interest of the Company, the above notice period of three Business Days may be waived by the unanimous consent of the directors expressed in writing.

- 24.2 Convening notices may validly be made in writing, or sent by electronic mail, provided that no notice (other than the resolution fixing their time) need be given as to regularly scheduled meetings. Meetings of the Board of Directors shall be convened at the registered office of the Company or at the place indicated in the notice convening the meeting.
- 24.3 The meetings of the Board of Directors shall be conducted under the chairpersonship of the chairperson of the Board of Directors or, in case of impediment, a vice-chairperson (if any has been elected) or a director appointed by his/her fellow directors.
- 24.4 Any or all of the directors may participate in a meeting of the Board of Directors by means of telephone, videoconference or similar communications equipment by means of which all persons participating in the meeting can hear each other. Participation in a meeting by such means shall constitute presence in person at such meeting.

Decisions of the Board of Directors may also be adopted, without any physical meeting, by the unanimous consent of the directors expressed in writing.

- 24.5 The Board of Directors may invite one or more individuals whether or not employees of the Company or its subsidiaries within the meaning of article 1:15 of the Code, to contribute their experience and knowledge to the deliberations of the Board of Directors and may, to that effect and for a duration that it determines for each of them, allow them to attend its meetings, in whole or in part, in an advisory and non-voting capacity. Such individuals will not qualify as director for the purpose of these Articles of Association, the Code or any other purpose. The Board of Directors determines the amount of their remuneration.
- 24.6 All directors shall be provided with all information that the Board of Directors considers necessary for the performance of its duties and all information that the Board of Directors considers is material to the Company.

Article 25. DELIBERATIONS

- 25.1 The Board of Directors may validly deliberate only if the majority of the directors are present or represented. As an exception, in any case of force majeure, the quorum for the Board of Directors shall be four directors present or represented and the Board of Directors may validly deliberate only with respect to actions required to be taken to protect the interests of the Company in connection with the circumstances of force majeure.
- 25.2 Any director may grant a proxy to another director in order to be represented at a specific meeting. Such proxies must be recorded in a proxy form bearing the director's signature (which may be a digital signature as defined in the Civil Code) and must be notified to the Board of Directors by letter, or e-mail. However, no director may hold a proxy for more than one director. A director so represented shall be deemed to be present in person.
- 25.3 Decisions at a meeting of the Board of Directors shall be taken by a majority of the votes cast, without taking into account abstentions. In case of an equality of votes, the chairperson of the meeting shall not have any casting vote.

Article 26. MINUTES OF BOARD MEETINGS

The decisions of the Board of Directors shall be recorded in minutes, kept at the registered office of the Company, and signed by the person acting as chairperson and a majority of members present at the meeting (including the chairperson).

Copies and extracts of the minutes under private deed may validly be signed by two directors, by a person to whom the day-to-day management of the Company has been delegated or by the Company Secretary.

Article 27. POWERS OF THE DIRECTORS

The Board of Directors shall have the powers to do all that is necessary or useful to achieve the corporate purpose of the Company, with the exception of those powers reserved to the Shareholders' Meeting by law or these Articles of Association.

Irrespective of the Board of Directors' general powers of representation as a collegial body, the Company shall be validly represented by two directors acting jointly.

Article 28. CONFLICTS OF INTEREST AND RELATED-PARTY TRANSACTIONS

Directors will be required to arrange their personal and business affairs so as to avoid conflicts of interest with the Company within the meaning of article 7:96 of the Code. Any director with a direct or indirect conflicting financial interest on any matter before the Board of Directors will be required to bring it to the attention of both the statutory auditor and fellow directors, will not take part in any deliberation and vote related thereto, and will not be taken into account for the purpose of calculating the quorum for the vote by the Board of Directors on such matter. Conflicts of interest within the meaning of article 7:96 of the Code will be disclosed in accordance with the relevant legal provisions.

Any proposed related party decision or transaction falling within the scope of article 7:97 of the Code shall be submitted to a committee of three independent directors in accordance with such article and shall only be entered into after review by such committee.

Article 29. EXECUTIVE COMMITTEE - DELEGATED POWERS

- 29.1 The Board of Directors may set up an executive committee from amongst or outside its members (the Executive Committee) which shall not constitute a management board within the meaning of articles 7:104 and 7:107 of the Code. It shall fix the powers and procedures thereof and set the remuneration of its members which shall be charged to overheads.
- 29.2 The Board of Directors may confer the powers of day-to-day management of the Company, together with the power to represent the Company for such day-to-day management, upon one or more persons who may but need not be directors.
- 29.3 The Board of Directors, the Executive Committee, and the persons with the powers of day-to-day management within the limits of those daily management powers, may likewise grant special and specific authority to one or more persons of their choice. The production of a copy of the decision of the Board of Directors or the authority conferred by one or more persons with the powers of day-to-day management, by the Executive Committee or agents holding special authority shall constitute sufficient proof of their powers.

Article 30. AUDIT

- 30.1 The financial position, annual accounts and compliance with the law and these Articles of Association of transactions required to be disclosed in the annual accounts shall be audited by one or more statutory auditors, who may be natural or legal persons, appointed by the Shareholders' Meeting.
- 30.2 The statutory auditors shall hold office for renewable periods of three years. The appointment of retiring auditors which have not been re-appointed shall terminate immediately after the closing of the ordinary Shareholders' Meeting.

Article 31. REMUNERATION – EMOLUMENTS

The Shareholders' Meeting may grant the directors' emoluments, which shall be chargeable to overheads.

The statutory auditor(s) shall be remunerated by a fixed fee determined by the Shareholders' Meeting at the beginning of his or her (their) mandate and which may be changed only by agreement between the parties.

The Company is authorised to deviate from the provisions of article 7:91, indents 1 and 2, of the Code, in respect of any persons falling within the scope of such provisions.

Article 32. SHAREHOLDERS' MEETINGS

- 32.1 The ordinary Shareholders' Meeting shall be held, each year, on the last Wednesday of April at 11:00 am Belgian time, in one of the municipalities of the Brussels-Capital Region, in Leuven or in Liège, at the place designated by the convening notice. If such day is a legal public holiday in Belgium, the meeting shall be held at the same hour on the following Business Day.
- 32.2 The Board of Directors or the statutory auditor of the Company will be able to convene any ad hoc, special or extraordinary Shareholders' Meeting. Shareholders representing 10% of the Company's capital will also be able to request the Board of Directors to convene a Shareholders' Meeting. The ad hoc, special or extraordinary Shareholders' Meetings shall be held on the day, at the hour and in the place designated by the convening notice. They may be held at locations other than the registered office.
- 32.3 The notice made by the Board of Directors may validly be signed by the chairperson or by a person to whom the day-to-day management has been delegated.

Article 33. ADMISSION TO SHAREHOLDERS' MEETINGS

- 33.1 In order to have the right to participate in and vote at the Shareholders' Meeting, shareholders must:
- (a) have the ownership of their Shares recorded in their name, as at 24:00 Belgian time on the fourteenth calendar day preceding the date of the Shareholders' Meeting (the Record Date):
- (i) through registration in the register of the registered Shares of the Company, for holders of registered Shares; or
- (ii) through book-entry in the accounts of an authorised account holder or clearing organisation, for holders of dematerialised Shares; and
- (b) notify the Company (or the person designated by the Company) by returning a signed original paper form or, if permitted by the Company in the notice convening the Shareholders' Meeting, by sending a form electronically (in which case the form shall be signed by means of an electronic signature in accordance with applicable Belgian law), at the latest on the sixth calendar day preceding the day of the Shareholders' Meeting, of their intention to participate in the Shareholders' Meeting, indicating the number of Shares in respect of which they intend to do so. In addition, the holders of dematerialised Shares must, at the latest on the same day, provide the Company (or the person designated by the Company), or arrange for the Company (or the person designated by the Company) to be provided, with an original certificate issued by an authorised account holder or a clearing organisation certifying the number of Shares owned on the Record Date by the relevant shareholder and for which it has notified its intention to participate in the Shareholders' Meeting.

An issuer of certificates representing registered Shares must notify its capacity as an issuer to the Company, which will record such capacity in the register of such Shares. An issuer who does not notify its capacity to the Company can only vote at a Shareholders' Meeting if the written notification indicating its intention to participate in that Shareholders' Meeting specifies its capacity of issuer.

An issuer of certificates representing dematerialised Shares must notify its capacity of issuer to the Company before exercising any vote, at the latest through the written notification indicating its intention to participate in the Shareholders' Meeting, failing which such Shares cannot participate in voting.

33.2 Any shareholder with the right to vote may either personally participate in the Shareholders' Meeting or give a proxy to another person, who need not be a shareholder, to represent it at a Shareholders' Meeting.

A shareholder may designate, for a given Shareholders' Meeting, only one person as proxy holder, except in circumstances where Belgian law allows the designation of multiple proxy holders.

The appointment of a proxy holder may take place in paper form or electronically (in which case the form shall be signed by means of an electronic signature in accordance with applicable Belgian law), through a form which shall be made available by the Company. The signed original paper form or electronic form must be received by the Company at the latest on the sixth calendar day preceding the date of the Shareholders' Meeting.

Any appointment of a proxy holder shall comply with relevant requirements of applicable Belgian law in terms of conflicting interests, record keeping and any other applicable requirement.

33.3 Prior to the Shareholders' Meeting, the shareholders or their proxies are required to sign an attendance sheet, indicating their first name, last name, and place of residence or corporate denomination and registered office, as well as the number of Shares in respect of which they are participating in the Shareholders' Meeting. Representatives of legal entities must provide the documents evidencing their capacity as bodies or special proxy holders.

The natural persons, shareholders, bodies or proxy holders who take part in the Shareholders' Meeting must be able to prove their identity.

33.4 The holders of profit sharing certificates, non-voting shares, subscription rights, convertible bonds as well as the holders of certificates issued with the assistance of the Company and representing securities issued by the latter, may insofar as the law entitles them to do so attend the Shareholders' Meeting, and, as the case may be, participate in voting. If they propose to participate, they are subject to the same formalities concerning admission and access, and forms and filing of proxies, as those imposed on the shareholders.

Article 34. COMPETING RIGHTS

Co-owners, as well as pledgors and pledgees, must be represented by a sole person. The usufructuaries will represent the bare owners unless otherwise provided in the deed establishing the usufruct or agreed upon. In the event of dispute between the bare owner and the usufructuary concerning the existence or scope of such agreement or provision, only the usufructuary shall be admitted to participate in the Shareholders' Meeting and participate in voting.

Article 35. REMOTE VOTING BEFORE THE SHAREHOLDERS' MEETING

Any shareholder may vote remotely before the Shareholders' Meeting, by sending a paper form made available by the Company or, if permitted by the Company in the notice convening the Shareholders' Meeting, by sending a form electronically (in which case the form shall be signed by means of an electronic signature in accordance with applicable Belgian law). The original signed paper form must be received by the Company at the latest on the sixth calendar day preceding the date of the Shareholders' Meeting. Voting through the sending of the signed electronic form may occur until the calendar day before the date of the Shareholders' Meeting.

The Company may also organise a remote vote before the Shareholders' Meeting through other electronic communication methods, such as, among others, through one or several websites. It shall specify the practical terms of any such remote vote in the convening notice.

The Company will ensure that, when arranging remote electronic voting before the Shareholders' Meeting, either through the sending electronically of a form or through other electronic communication methods, the Company is able, through the system used, to control the identity and capacity as shareholder of each person casting a vote electronically.

Shareholders voting remotely, must, in order for their vote to be taken into account for the calculation of the quorum and voting majority, comply with the conditions set out in Article 33.

Article 36. CHAIRPERSONSHIP AND OFFICE

The Shareholders' Meeting shall be chaired by the chairperson of the Board of Directors, or, in case of absence or impediment, by a vice-chairperson, or in the absence of all such, by a director previously appointed for this purpose by the Board of Directors, or, in the absence of such appointment, by any of the directors present.

The chairperson of the meeting shall appoint the secretary, who does not need to be a shareholder. If the number of participants so requires, he or she shall appoint two tellers from among the shareholders or their representatives. The chairperson, the secretary and the tellers together form the office of the Shareholders' Meeting.

The chairperson can appoint the office of the meeting prior to the opening of the meeting, and the latter can proceed to the verification of the powers of the participants prior to this opening.

Article 37. AGENDA AND DELIBERATIONS

- 37.1 The Shareholders' Meeting may deliberate only the items on its agenda.
- 37.2 One or more shareholders representing at least 3% of the capital of the Company may request for items to be added to the agenda and submit resolution proposals in relation to existing agenda items or new items to be added to the agenda provided that they evidence the holding of such 3% shareholding as at the date of their request by a certificate evidencing the registration of the Shares in the register of Shares of the Company with respect to registered Shares or by a certificate issued by an authorised account holder or a clearing organisation certifying the book-entry of the Shares in one or more accounts held by such account holder or clearing organisation, with respect to dematerialised Shares.

Such right shall not be available in relation to a second extraordinary Shareholders' Meeting that is convened for lack of a quorum at the first extraordinary Shareholders' Meeting.

The new agenda items and/or resolution proposals should be received by the Company in signed original paper form or electronically (in which case the form shall be signed by means of an electronic signature in accordance with applicable Belgian law), at the latest on the twenty-second calendar day preceding the date of the Shareholders' Meeting and the Company shall publish a revised agenda at the latest on the fifteenth calendar day preceding the date of the Shareholders' Meeting.

The handling of such new agenda items and/or resolution proposals during the Shareholders' Meeting is subject to the relevant shareholder(s) having satisfied, with respect to Shares representing at least 3% of the capital, the conditions set forth in Article 33.

37.3 Each Share gives its holder the right to one vote at Shareholders' Meetings.

For all matters, except as otherwise provided in these Articles of Association or by law, resolutions shall be passed by a majority of the votes cast.

Voting shall be by show of hands or through electronic devices.

37.4 Shareholders' Meetings may be broadcast by way of live or recorded video conferences or audio conferences, in part or as a whole, via one or more websites as the case may be, from the place where the meeting is held to one or more remote places where some people, identified or not, are located. Physical persons who attend a Shareholders' Meeting agree by this very fact that their picture may be so transmitted.

Article 38. MINUTES OF THE SHAREHOLDERS' MEETING

The minutes of Shareholders' Meetings shall be signed by the chairperson of the Shareholders' Meeting, the secretary, the tellers and such shareholders (or proxies) who wish to do so.

Copies and extracts of the minutes under private deed may validly be signed by two directors, by a person to whom the day-to-day management of the Company has been delegated or by the Company Secretary.

Article 39. ADJOURNMENTS

39.1 Irrespective of the items on the agenda, the Board of Directors may adjourn any ordinary or other Shareholders' Meeting. It can use this right at any time, but only after the opening of the meeting. Its decision, which does not have to be justified, must be notified to the Shareholders' Meeting before the end of the meeting, and mentioned in the minutes.

Such adjournment cancels all decisions taken during the Shareholders' Meeting.

39.2 The adjourned Shareholders' Meeting shall be held again within five weeks with the same agenda. Shareholders wishing to participate in such Shareholders' Meeting shall fulfil the admission conditions set out in Article 33.1(a) and (b). To this effect, a record date shall be set on the fourteenth calendar day at 24:00 Belgian Time preceding the date of the second Shareholders' Meeting.

Article 40. POWERS OF THE SHAREHOLDERS' MEETING

A duly constituted Shareholders' Meeting represents all the shareholders.

The Shareholders' Meeting has the powers conferred on it by law. In addition, any acquisition or disposal of tangible assets by the Company for an amount higher than the value of one third of the Company's consolidated total assets as reported in the Company's most recent audited consolidated financial statements shall be within the exclusive jurisdiction of the Shareholders' Meeting and shall be adopted with a majority of at least 75% of the votes cast at the Shareholders' Meeting, regardless of the number of Shares attending or represented.

Article 41. TRANSACTIONS WITH A MAJOR SHAREHOLDER

In the event of (i) a contribution in kind to the Company with assets owned by any person or entity which is required to file a transparency declaration pursuant to applicable Belgian law or a subsidiary (within the meaning of article 1:15 of the Code) of such person or entity, or (ii) a merger of the Company with such a person or entity or a subsidiary of such person or entity, then such person or entity and its subsidiaries shall not be entitled to vote on the resolution submitted to the Shareholders' Meeting to approve such contribution in kind or merger.

Article 42. ACCOUNTING RECORDS

The financial year shall begin on the first of January and end on the thirty-first of December each year.

At the end of each financial year, the Board of Directors shall draw up an inventory and the annual accounts of the Company.

Article 43. PROFIT ALLOCATION

The Shareholders' Meeting shall allocate the net profit on the recommendation of the Board of Directors, provided that no less than 5% of the net profits of the Company, after deduction of overheads and depreciation, shall be allocated each year to the legal reserve. Such allocation to the legal reserve shall cease to be compulsory once the legal reserve has become equal to one-tenth of the share capital.

Ordinary Shares and Restricted Shares will have the same rights in relation to dividends and other distributions.

Article 44. PAYMENT OF DIVIDENDS

The annual dividends shall be paid at the dates and places decided by the Board of Directors.

The Board of Directors may pay an interim dividend in accordance with article 7:213 of the Code.

Article 45. WINDING UP

If the Company is dissolved, it shall be wound up in the manner decided by the Shareholders' Meeting, which shall appoint the liquidator(s).

The liquidator(s) shall have all powers set out under the Code, subject to any restrictions imposed by the Shareholders' Meeting.

After all liabilities have been cleared, the balance of the assets owned by the Company shall be distributed equally among all the Shares.

Article 46. BONDHOLDERS' MEETINGS

General meetings of bondholders shall be held in accordance with the provisions of articles 7:161 to 7:176 of the Code. The office for general meetings of bondholders shall be composed as provided for in Article 36.

Article 47. ADDRESS FOR SERVICE

Without prejudice to the next indent, any shareholder, bondholder, director, statutory auditor or liquidator of the Company not residing in Belgium shall elect an official address in Belgium. Otherwise he or she shall be deemed to have elected the registered office of the Company as his or her official address where all communications, notices, processes and documents may validly be sent to or served upon him or her.

Any shareholder, security holder, holder of certificate issued with the assistance of the Company, director or statutory auditor may communicate to the Company an email address at which such person can be reached. Any communication to such email address by the Company will be deemed valid."

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,519,988,309
AGAINST	798,762
ABSTENTION	794,628

The resolution was consequently approved.

ORDINARY SHAREHOLDERS' MEETING

ACKNOWLEDGMENT OF REPORTS

The chairman asks the meeting to approve that it be recorded in the minutes that the reading of the documents listed below was dispensed with as the shareholders were sufficiently familiar with them:

- 1/ Management report by the Board of Directors on the accounting year ended on 31 December 2019.
 - 2/ Report by the statutory auditor on the accounting year ended on 31 December 2019.
 - 3/ The consolidated annual accounts on the accounting year ended on 31 December 2019.
 - 4/ The statutory annual accounts on the accounting year ended on 31 December 2019.

FOURTH RESOLUTION: Approval of the statutory annual accounts.

The chairman submits to the meeting the proposal to approve the statutory annual accounts relating to the accounting year ended on 31 December 2019, including the following allocation of the result:

		<u>000, EUR</u>
Profit of the accounting year:	+	10,404,860
Profit carried forward:	+	12,604,362
Result to be allocated:	=	23,009,222
Transfer from reserves:	+	0

Deduction for the unavailable reserve:
- 791,390
Gross dividend for the shares (*):
- 2,556,697
Balance of carried forward profit:
= 19,661,136

(*) On a per share basis, this represents a gross dividend for 2019 of EUR 1.30.

Taking into account the gross interim dividend of EUR 0.80 per share paid in November 2019, a balance gross amount of EUR 0.50 will be payable as from 11 June 2020 (JSE: 15 June 2020), i.e. a balance dividend net of Belgian withholding tax of EUR 0.35 per share (in case of 30% Belgian withholding tax) and of EUR 0.50 per share (in case of exemption from Belgian withholding tax).

The actual gross dividend amount (and, subsequently, the balance amount) may fluctuate depending on possible changes in the number of own shares held by the Company on the dividend payment date.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,517,431,733
AGAINST	3,209,230
ABSTENTION	940,736

The resolution was consequently approved.

FIFTH RESOLUTION: Discharge to the Directors.

The chairman submits to the meeting the proposal to grant discharge to the Directors for the performance of their duties during the accounting year ended on 31 December 2019.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,502,566,796
AGAINST	16,551,471
ABSTENTION	2,463,432

The resolution was consequently approved.

SIXTH RESOLUTION: Discharge to the statutory auditor.

The chairman submits to the meeting the proposal to grant discharge to the statutory auditor for the performance of his duty during the accounting year ended on 31 December 2019.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,504,845,737
AGAINST	11,457,103
ABSTENTION	5,278,859

The resolution was consequently approved.

SEVENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Board of Directors, renew the appointment of Ms. **BURNS Michele** as independent director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Ms. Burns expressly stated and the Board is of the opinion that she complies with the functional, family and financial criteria of independence as provided for in the 2020 Belgian Corporate Governance Code.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,478,670,774
AGAINST	41,845,574
ABSTENTION	1.065.351

The resolution was consequently approved.

EIGHT RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Board of Directors, renew the appointment of Mr. **SCETI Elio Leoni** as independent director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Mr. Sceti expressly stated and the Board is of the opinion that he complies with the functional, family and financial criteria of independence as provided for in the 2020 Belgian Corporate Governance Code.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,462,776,875
AGAINST	57,753,257
ABSTENTION	1,051,567

The resolution was consequently approved.

NINTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Reference Shareholder, renew the appointment as director of Mr. **VAN DAMME Alexandre**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,267,284,506
AGAINST	252,332,969
ABSTENTION	1,964,224

The resolution was consequently approved.

TENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Reference Shareholder, renew the appointment as director of Mr. **DE SPOELBERCH Grégoire**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,309,942,840
AGAINST	209,675,900
ABSTENTION	1,962,959

The resolution was consequently approved.

ELEVENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Reference Shareholder, renew the appointment as director of Mr. **CORNET DE WAYS RUART Paul**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,256,437,953
AGAINST	263,178,831
ABSTENTION	1,964,915

The resolution was consequently approved.

TWELFTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Reference Shareholder, renew the appointment as director of Mr. **LEMANN Paulo**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,308,445,380
AGAINST	211,020,656

ABSTENTION	2,115,663
------------	-----------

The resolution was consequently approved.

THIRTEENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Reference Shareholder, renew the appointment as director of Ms. **ASUNCION ARAMBURUZABALA María**, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,295,481,542
AGAINST	223,985,643
ABSTENTION	2,114,514

The resolution was consequently approved.

FOURTEENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to acknowledge the end of the mandate of Mr. HERRMANN TELLES Marcel as director and, upon proposal from the Reference Shareholder, appoint Mr. THOMPSON MOTTA Roberto as director, for a period of four years ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2023. Mr. Roberto Thompson Motta, a Brazilian citizen, received a BS in Mechanical Engineering from Pontifícia Universidade Católica do Rio de Janeiro and an MBA from The Wharton School of the University of Pennsylvania. He is a co-founder and member of the Investment Committee of 3G Capital, a global investment firm headquartered in New York. Mr. Thompson served on the Board of Directors of AB InBev from 2004 until 2014 and has served since 2001 on the Board of Directors of AmBev S.A. where he is a member of the Operational and Finance Committee. Mr. Thompson has served on the Board of Directors of Restaurant Brands International since 2013, Lojas Americanas S.A. since 2001, São Carlos Empreendimentos e Participações S.A. since 2001 and StoneCo Ltd., a leading provider of financial technology solutions, since 2018, where he chairs the Finance Committee. From 1993 to 2004, he was one of the founding partners of GP Investments Ltd. and a member of its Board of Directors until 2010. Mr. Thompson is a member of the Academy of the University of Pennsylvania, The International Council of The Metropolitan Museum of Art in New York and a Patron of the Museum of Modern Art of São Paulo.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,259,453,613
AGAINST	259,973,503
ABSTENTION	2,154,583

The resolution was consequently approved.

FIFTEENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Restricted Shareholders, renew the appointment as Restricted Share Director of Mr. **BARRINGTON Martin J.** for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,266,655,076
AGAINST	251,979,497
ABSTENTION	2,947,126

The resolution was consequently approved.

SIXTEENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Restricted Shareholders, renew the appointment as Restricted Share Director of Mr. **GIFFORD William F., Jr.**, for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,311,184,667
AGAINST	208,213,529
ABSTENTION	2,183,503

The resolution was consequently approved.

SEVENTEENTH RESOLUTION: Appointment of Directors.

The chairman submits to the meeting the proposal to, upon proposal from the Restricted Shareholders, renew the appointment as Restricted Share Director of Mr. **SANTO DOMINGO DÁVILA Alejandro**, for a period of one year ending at the end of the shareholders' meeting which will be asked to approve the accounts for the year 2020.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

OT WITHCIT	
FOR	1,297,499,366
AGAINST	221,936,751
ABSTENTION	2,145,582

The resolution was consequently approved.

For good order, the secretary declares that all current directors of the Company (including those whose mandate has not been renewed at the present general meeting) also elect domicile at the Company's registered office.

**

EIGHTEENTH RESOLUTION: Remuneration Policy and Remuneration Report of the Company.

The chairman submits to the meeting the proposal to approve the Remuneration Report for the financial year 2019 as set out in the 2019 annual report, including the remuneration policy. The 2019 annual report and the remuneration report comprising the remuneration policy are available on the Company's website.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,219,371,500
AGAINST	299,966,679
ABSTENTION	2,243,520

The resolution was consequently approved.

NINETEENTH RESOLUTION: Powers.

The chairman submits to the meeting the proposal to grant, without prejudice to other delegations of powers to the extent applicable, powers to Mr. Jan Vandermeersch, *Global Legal Director Corporate*, with power to substitute, to proceed to (i) the signing of the restated articles of association and their filings with the clerk's office of the Enterprise Court of Brussels as a result of the approval of the resolutions under item 1 above, and (ii) any other filings and publication formalities in relation to the above resolutions.

Vote:

The proposal is put to the vote. It is adopted as follows:

- 1/ Number of shares for which votes are validly cast: 1,521,581,699
- 2/ Percentage that the aforementioned number of shares represent in the capital: 75.35
- 3/ Number of votes validly cast: 1,521,581,699

Of which

FOR	1,519,781,939
AGAINST	793,436
ABSTENTION	1,006,324

The resolution was consequently approved.

CLOSING OF THE MEETING

The meeting is closed.

DOCUMENT DUTIES

The duty amounts to EUR 95.00.

IDENTITY

The notary confirms the surname, first names, date and place of birth and domicile of the secretary of the meeting on presentation of his identity card.

AS RECORDED BY THESE MINUTES

Drawn up on the date and place indicated above.

After full reading of these minutes, partly by the chairman and partly in English by Notary Carnewal, the minutes are signed by the secretary and the tellers and by me, Mr. Missoul, Notary.

Annual General Meeting 2020 O&A

First question

"Since acquiring my shares in NEWBELCO there has been a steady decline in the dividends paid out by the company in Euros per share. In 2019, shareholders received only 11% of the profit available for allocation as Dividends. This has been accompanied by a correspondingly steady decline in the share price. Could the Directors please indicate by when shareholders might start receiving Dividends more in line with the capital risk to which they are exposed in holding the Company's shares?"

(Martin J. Barrington - Chairman)

First, we heard Brito explain in his presentation the company's four capital allocation priorities and use of cash. In brief, the first priority is to invest in our brands and growth opportunities; the second is to deleverage; the third is, subject to strict discipline and the deleveraging commitment, to look at opportunistic M&A. And then the fourth is to return excess cash to shareholders in the form of dividends or share buybacks.

Second, Brito also has explained the company's decision to propose a final dividend of 50 Euro cents.

Finally, you'll remember that in March 2020 the company withdrew its previously-issued 2020 outlook, including with respect to dividends. That was because of the uncertainty, volatility and fast-moving developments of the COVID-19 pandemic in the markets where we operate. And the company has not provided new guidance at this time.

The company continues to carefully monitor developments in this complex and constantly-evolving environment, which will inform any future decisions, including on dividend payments.

Second question

"Premiumization strategy: A year ago or so AB InBev emphasized on the fact that there is still a significant premiumization potential for beer. In the environment post Covid, do you think the premiumization strategy can still be well executed without impacting your market share?"

(Carlos Brito - CEO)

Premiumization remains a top growth opportunity for our company today and in the long-term. It continues to be a trend across our markets, and it is a critical driver of growth as it generates incremental top and bottom-line. Evidence of this is that our High End portfolio and global brands continue to grow ahead of our core portfolio.

In the past, we have seen that beer tends to show less elasticity in soft economic environments as it is considered an affordable luxury in many markets. In addition, compared to wine and spirits, the beer category is in its early stages of premiumization and this provides an opportunity for us to grow and close the gap.

We are well positioned to capture this opportunity by leveraging our market maturity model and our category expansion framework. If you're interested in learning more about our market maturity model and category expansion framework, please refer to the investor section on our website and to our Annual Report.

Third question

"Becoming a more consumer-centric organization: Many believe that AB InBev in the past focused too much on being a production driven organization, so much so that it lost touch with fast changing customer preferences. Could you tell us what concrete steps have been taken now to evolve AB InBev into a more customer centric organization and where are we in this transition journey?"

(Carlos Brito - CEO)

Consumer-centricity and innovation drive all we do. We have increased focus on these drivers during the past five years and taken steps to shape our organization to be more agile and consumer-centric than ever, which is critical to address quickly evolving consumer needs.

For example, in 2015 we created ZX Ventures, a business unit tasked with accelerating innovation by addressing emerging consumer needs with new and disruptive products and businesses. Some major product innovations and businesses that have come out of ZX Ventures include:

- O Drinkworks, an at-home alcohol system developed in partnership with Keurig that utilizes pods to make single-serve ready-to-drink cocktails,
- o our owned direct-to-consumer e-commerce platforms in several countries, and
- o our strengthened specialty portfolio of premium craft and specialty beers.

ZX Ventures accounted for more than 15% of our revenue growth by the end of 2019.

In 2018, we created two new senior leadership positions to capture organic growth opportunities in non-alcohol beverages and direct-to-consumer initiatives.

We also expanded draftLine, our internal agency that creates consumer insights by data analytics, so we can swiftly respond with new products and campaigns.

Finally, this year COVID-19 has tested our agility and consumer-centric approach. In today's environment, consumers are changing their habits very fast, increasing the adoption of eCommerce and contactless delivery. To respond, we have accelerated our business transformation enabled by technology, which was well underway before the pandemic hit.

We believe that the strong foundations we have built in the past few years, put us at an advantage to continue driving future growth within these evolving trends.

Fourth question

"Growth market exposure: Almost 2/3rd of your volumes and about similar proportion of your operational cash flow comes from emerging markets. The macro economic environment for these geographies is expected to be more negatively impacted by Covid and could take much longer to recover. What steps is the company taking to prepare for a delayed recovery in the growth markets? And linked to the emerging markets, to what extent you can manage the big currency devaluations? Could you give an estimate what the impact of currency fluctuations would be for 2020 on revenues, EBITDA and net profit if these exchange rates would remain at the current levels?"

(Carlos Brito - CEO)

We are proud to have a diverse geographic footprint with 10 markets contributing to 80% of our EBITDA, reducing our exposure to any given country. We have a long history of operating in emerging markets and recognize that they are inherently volatile. Our ownership culture requires us to take a long-term view of our business and weather such volatility to pursue the growth opportunities that are also characteristic of these markets. We believe we have the right people, portfolio and strategy in place to deliver long-term, sustainable growth and expand the global beer category.

We have not provided guidance on the impact of currency fluctuations on our 2020 results. However, what I can say with respect to FX volatility is that our hedging policy remains unchanged - we continue to hedge our FX and commodity exposures when possible on a 12 month forward rolling basis.

Fifth question

"Deleveraging efforts: With the impact of Covid delaying your deleveraging efforts from organic cash flows, it might be a while for AB InBev to reach your stated leverage level (of 2x net debt/ebitda). What other options is the group considering to reduce leverage?"

(Fernando Tennenbaum - CFO)

As discussed previously, our capital allocation priorities are the following:

- The first priority for the use of cash is to invest behind our brands and to take full advantage of the organic growth opportunities in our business.
- o Second, deleveraging to around a 2x net debt to EBITDA ratio remains our commitment, and we will prioritize debt repayment in order to meet this objective.
- Third, with respect to M&A, we will always be ready to look at opportunities when and if they arise, subject to our strict financial discipline and deleveraging commitment.
- Our fourth priority is returning excess cash to shareholders in the form of dividends and/or share buybacks.

With this being said, we must exercise prudent measures during times of uncertainty and volatility, including efficient management of discretionary expenditures, especially those which may not prove effective in the current environment.

We have also taken a decision to reduce the amount of the proposed final 2019 dividend, given the uncertainty, volatility and continued impact of the COVID-19 pandemic. We believe this is prudent and in the best interests of the Company.

This decision is consistent with our financial discipline, deleveraging commitments and other actions taken to navigate this environment.

Additionally, we are always reviewing our asset base to identify non-core assets that can be divested as part of our normal business operations. At this stage, there are no major divestments plans to highlight.

Sixth question

"Sustainability: As world's largest beer company, we believe AB InBev has far greater responsibility in promoting responsible drinking and in this regards your sustainability targets pay relatively less attention on this aspect. Could you share with us what targets have been placed to promote even further when it comes to creating awareness on "Responsible drinking"?"

(Carlos Brito - CEO)

We want every experience with beer to be a positive one.

We believe that harmful consumption of alcohol is bad for our consumers, our colleagues, our families, our communities, and our business. That is why we take seriously our responsibility to help reduce and prevent the harmful use of alcohol across the world.

We have promoted responsible drinking for decades and, to support our commitment, in 2015 we set our Global Smart Drinking Goals to be achieved by 2025, which you can find on our website and in our Annual Report.

By sharing what we learn and collaborating with public health authorities, trade associations, experts and civil society, we know AB InBev can be part of the solution.

The Global Smart Drinking Goals are different from and in addition to our 2025 Sustainability Goals, which cover environmental sustainability and resilience of our value chain and our communities. However, both sets of goals are connected to our aspiration of building a company to last for the next one hundred years and beyond.